Author’s response to reviews

Title: Misoprostol for the prevention of post-partum haemorrhage in Mozambique: An analysis of the interface between maternal health, human rights and development

Authors:

Karen Hobday (karen.hobday@menzies.edu.au)

Anthony Zwi (a.zwi@unsw.edu.au)

Caroline Homer (caroline.homer@burnet.edu.au)

Renae Kirkham (Renae.Kirkham@menzies.edu.au)

Jennifer Hulme (jennifer.hulme@uhn.ca)

Páscoa Wate (pascoawate@yahoo.com.br)

Ndola Prata (ndola@berkeley.edu)

Version: 2 Date: 16 Feb 2020

Author’s response to reviews:

Dear Editor,

We have made the following changes as requested:

1. Verbal consent: this statement has been added to the manuscript in the ethics declaration statement:

Pg 28, Declarations section, Line 23:

‘Approval was granted to obtain written and verbal consent from the international and local ethics committees’.

2. Figure 1

The permissions obtained have been indicated in figure 1 legend:

Figure 1. The Health, Human Rights and Development Triangle (Originally printed in Tarantola et al., 2008 (20) with permission obtained to re-print. This figure has been republished in Tarantola et al., 2013 (21)).
3. Figure 2

Figure 2 is the first author’s work. This has been indicated in figure 2 legend:

Figure 2. The Evolution of the Distribution of Misoprostol for Prevention of Postpartum Hemorrhage in Mozambique (Figure developed by K. Hobday and was published in Hobday et al, 2019 (22) which is an open access journal).

Please let me know if you have any questions or require any further changes.

Best regards

Karen Hobday